 
CENWP-OP-B                                                                                                                               13 August, 2001

MEMORANDUM FOR CENWP-OP, ATTN: Gary Johnson

SUBJECT: Bonneville Lock and Dam, Fishway and Fish Activities for week 32 of 2001, which covers the period 05 through 11 August.
1.  OPERATION SUMMARY.

1.1 Daily river flow averaged 83.0 to 126.2kcfs.  Daily spill averaged 9.1 to 49.6kcfs.  Daily project tailwater elevation averaged 8.3’ to 11.5’ msl.  Daily project forebay elevation averaged 74.9’ to 75.7’ msl.  River temperature averaged 68 to 70° F.  

1.2 Spill bays 1 and 18, which supply attraction flow for Cascades Island and Bradford Island ladders respectively, remain open 4”.  Night spill (2000 until 0100) of 45 kcfs began on 24 July.  24 hour spill of 50kcfs began at 1400 on 09 August and will continue until further notice.

1.3 Unit 3 remains out of service for turbine rehab.  Unit 17 remains out of service for its four-year overhaul.  Unit 5 remains out of service for turbine rehab.  On 06 August, Units 12-14 were out of service for transducer removal, all were back in service by 1436.  On 07 August, Units 11, 15, 16, and 18 were closed for transducer removal and trash raking, all were back in service by 1630.  On 08 August, Unit 15 was out of service from 0936 until 1130 for transducer removal. 

1.4 Second Powerhouse fish units remain in service.  They provide attraction flow for the Washington Shore fish ladder.  The units are placed on standby from approximately 0000-0300 nightly if drawdown exceeds 1.0’.   

2.  ACTIVITIES. 
2.1 AWS Closures. FV4-3 and FV4-4 at B-branch and FV5-3 and FV5-4 are all operating in manual until the new computer system is calibrated.   FV3-9 and the Bradford Island exit were closed for cleaning from 1527 until 1540 on 06 August, from 1535 until 1545 on 09 August, and from 1000 until 1020 on 11 August.

2.2 STS/VBS Inspections.  No STSs were video inspected this reporting week.

2.3 Dewatering and Fish Salvage.  No dewatering or fish salvage this reporting week.

3.  RESEARCH.

3.1 WDFW Fish Counting.  WDFW continues 16-hour visual counts and 8-hour video counts at Bradford Island and Washington Shore ladders.

3.2 NMFS Saltwater rearing. The Saltwater rearing facility is in operation.  Sampling May- November.

3.3 PSMFC Adult PIT.  Looking at A and B branch and Cascade Island for noise interference.  Analyzing data collected at the Washington Shore ladder.

3.4 PSMFC Smolt Monitoring Program.  Collection continues 24h/d at the monitoring facility.  All flat plate monitoring at DSM1 ended at 0800 on 02 August.

3.5 WES Hydroacoustic Monitoring.  Equipment removal scheduled 30 July – 9 August.

3.6 UI Barging Condition and Separation By Code study.  Fall Chinook separation by code study runs 15 July through 17 September. 

3.7 USGS Juvenile Radiotracking.  Equipment removal in progress.

3.8 UI Adult Fallback. UI Adult Radiotracking/Tagging.  PCC entrance gates at the first powerhouse are being operated according to UI request until 31 October.  Tagging of Chinook began at the AFC&MF 2 April and will continue seven days a week until the end of September.   Researchers are adding ice to the anesthetic tank when water temperatures approach 70° F.

3.9 CRITFC/ODFW/WDFW Adult Trapping/Tagging.  Tagging Chinook and steelhead on Tuesdays and Thursdays.

3.10 NMFS Lamprey.  Lamprey are tagged and tracked throughout the week.  Spillway ladder entrances manually operated at night for a .5’ differential.  NMFS is collecting fish for USGS.

3.11 YIN Northern Pikeminnow Dam Angling.  Fishing 25 June – 30 September.

3.12 WDFW/ODFW Fall Chinook Counting.  Observations of the upriver fall Chinook begin 14 August.

4.  FISHWAYS. 
4.1 Project biologists inspected 05 through 11 August this reporting week.  

4.2 Fish Passage Plan Observations.

DATE
LOCATion
FPP Deviation
CAUSE
RESPONSE
TR/WAR

All week
PH2 collection channel electronic velocity meter.
PH2 is not yet working.
Not fully installed.
Working on final installation.


All week
FG3-28
Out of service
Manually closed because of displaced diffuser valve cover.
New covers are being designed, and will be installed during winter maintenance.
20013904

All week
FG3-27
Not in correct position for TW elev.
FG3-27 is open to replace FG3-28.
No action taken.


8/7-11
PH1 Collection channel velocity
Velocity meter registers less than 1.5 fps.
Water backs up in the channel during OG/SG tests.  
Monitoring situation.


All week
Main Dam Avian wires
Wires removed.
Spillway deflector installation.
Wire grid to be replaced in Mid-April.


8/11
FG3-26
Not in correct position for TW elev.
Parameters not yet set in the computer.
Operators closed diffuser manually.


8/6, 8/8
A branch water over weir
Staff gauge reads >1.3’
Unknown
B-branch reads correctly.


8/8
B-Branch Entrance gates.
Not in correct position for TW elev.
Parameters not yet set in the computer.
Gates manually adjusted.


8/11
FG6-13
Not in correct position for TW elev.
Parameters not yet set in the computer.
Operators closed diffuser manually.


4.3 Fish Ladder Temperatures.  Temperatures are measured 3 weirs down from the count station at Bradford Island and at the junction of the UMT and the main ladder at Washington shore.

Date
Wash. Shore
Bradford Is.

8/5
20.2
20.1

8/6
20.6
20.5

8/7
21.0
20.8

8/8
21.3
21.1

8/9
21.3
21.2

8/10
21.7
21.5

8/11
21.8
21.7

4.4 Sensor Calibration.  Water level sensors were checked for calibration on 10 August.

4.5 Zebra Mussel Monitoring.  The zebra mussel monitoring station was inspected and showed no signs of colonization this reporting week.  On 31 July, we received from BPA, a report of two zebra mussel “sightings” at Grand Coulee.

5. MISCELLANEOUS

5.1 PH2 JBS injury/mortality issue: Juvenile mortality remains low at the JMF, according to PSMFC reports.  

5.2 Because of low tailwater elevations and the physical limits of the weir gates, entrance openings for PH1 Collection Channel are less than 8.0’.   Gate 64 has been reprogrammed to maintain at least an 8’ opening.  

5.3 Closed OG/SG testing causes low differential at the north end of PH1 collection channel.  FV1-1 normally controls south channel differential, but is now reprogrammed to maintain north channel differential.  This change should also help maintain proper collection channel velocity during OG/SG tests.  

5.4 The B-Branch and Cascade Island computers were installed.  Sensors still need calibrated and some parameters are not set quite right yet.  Until then, entrance gates and FV4-3, 4-4, 5-3, and 5-4 are operated in manual.  

5.5 Spillway dives were conducted 06 August through 09 August, as part of the survey work for spill deflector construction.  Fishways were not taken out of criteria for these dives.   

                                                                                                                DEBORAH L. CHENOWETH

                                 

                                                             Operations Project Manager
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